Author's response to reviews

Title: The effectiveness of pelvic floor muscle training alone and with other physical therapies for the treatment of stress urinary incontinence in women: a systematic review

Authors:

Patricia B Neumann (cpneumann@ozemail.com.au)
Karen A Grimmer (Karen.grimmer@unisa.edu.au)
Yamini Deenadayalan (Yamini.deenadayalan@unisa.edu.au)

Version: 5 Date: 19 April 2006

Author's response to reviews: see over
Details of changes made to document:

Page 6, line 19 – Sentence added with additional reference
Observational studies provide the opportunity to establish the effectiveness of such interventions in routine clinical practice [18]. This is difficult to achieve in randomized trials [18] other than pragmatic trials [Roland and Torgerson]

Reference added: page 22, line 11
The use of outcome measures which are valid, reliable and sensitive to change is vital when considering the effects of treatment in order to detect valid changes which are greater than measurement error [Streiner, 1995]

Other minor editing changes have been made to correct labelling, reference list, as requested under Minor Essential Revisions